Author’s response to reviews

Title: Methodological challenges in European ethics approvals for a genetic epidemiology study in critically ill patients: the GenOSept experience

Authors:

Ascanio Tridente (ascanio.tridente@doctors.org.uk)
Paul Holloway (paul.holloway@imperial.ac.uk)
Paula Hutton (icugeneticresearch@orh.nhs.uk)
Anthony Gordon (anthony.gordon@imperial.ac.uk)
Gary Mills (g.h.mills@sheffield.ac.uk)
Geraldine Clarke (Geraldine.Clarke@health.org.uk)
Jean-Daniel Chiche (jean-daniel.chiche@cch.aphp.fr)
Frank Stueber (frank.stueber@insel.ch)
Christopher Garrard (cgarrard@mac.com)
Charles Hinds (c.j.hinds@qmul.ac.uk)
Julian Bion (j.f.bion@bham.ac.uk)
GenOSept Investigators (research@esicm.org)

Version: 3 Date: 18 Apr 2019

Author’s response to reviews:

Cover letter to accompany submission to BMC Medical Ethics following minor revisions

BMC Medical Ethics: METH-D-18-00146R2

Title: Methodological challenges in European ethics approvals for a genetic epidemiology study in critically ill patients: the GenOSept experience

Authors: Ascanio Tridente; Paul Holloway; Paula Hutton; Anthony Gordon; Gary Mills; Geraldine Clarke; Jean-Daniel Chiche; Frank Stueber; Christopher Garrard; Charles Hinds; Julian Bion
Dear Dr Eleanor Harding,

We are grateful for the possibility to further revise our manuscript (METH-D-18-00146R2), in light of the comments received. We have addressed the editor’s and reviewers’ comments.

1) We have removed all responses to reviewers

2) We have removed one of the two funding declarations

A “clean” version only is provided, as requested. We confirm that the manuscript has not been submitted, accepted for publication or published elsewhere.

Kind Regards

Ascanio Tridente

Ascanio Tridente
Consultant in Intensive Care
Whiston Hospital, UK